**Author's response to reviews**

**Title:** Factors influencing general practitioners in the referral of elderly cancer patients

**Authors:**

Fleur Delva (fleur.delva@isped.u-bordeaux2.fr)
Emilie Marien (e-marien@hotmail.fr)
Marianne Fonck (fonck@bergonie.org)
Muriel Rainfray (muriel.rainfray@chu-bordeaux.fr)
Jean-Louis Demeaux (jeanlouis.demeaux@wanadoo.fr)
Philippe Moreaud (p.moreaud@urmla.org)
Pierre Soubeyran (soubeyran_p@bergonie.org)
Annie J Sasco (annie.sasco@isped.u-bordeaux2.fr)
Simone Mathoulin-Pelissier (mathoulin@bergonie.org)

**Version:** 3 **Date:** 29 December 2010

**Author's response to reviews:**

BMC cancer; 2010 December 28: Revisions

Factors influencing general practitioners in the referral of elderly cancer patients

1) Modifications in the list of references


Ref 13 and 14 were errors and removed in the text as in the list of references

All references were now with new number after the reference 12, i.e ref 15 is now reference 13 , ref 16: now ref 14; ref 17: now ref 15….

2) We added a section “additional files” in the text, at the end of the manuscript
3) We formatted author's contributions into a paragraph

4) We formatted abbreviation into a paragraph